Author's response to reviews

Title: The impact of a computerized decision aid on empowering pregnant women for choosing vaginal versus cesarean section delivery - Study protocol for a randomized controlled trial

Authors:

Saeid Eslami (EslamiS@mums.ac.ir)
Azam Aslani (azamaslani@yahoo.com)
Fatemeh Tara (tara@mums.ac.ir)
Leila ghalighi (leila.ghalichi@gmail.com)
Fatemeh Erfanian (ErfanianF@mums.ac.ir)
Ameen Abu-Hanna (a.abu-hanna@amc.uva.nl)

Version: 5 Date: 23 June 2015

Author's response to reviews:

Dear Editor,

Thank you for the opportunity to resubmit our article "The impact of a computerized decision aid on vaginal versus cesarean section delivery - Study protocol for a randomized controlled trial" by Saeid Eslami, Azam Aslani, Fatemeh Tara, Lila Ghalighi, Fatemeh Erfanian, Ameen Abu-Hanna. Enclosed please find the re-submission for possible publication in Trials.

We are also grateful for the useful remarks made by you and the reviewer. We have addressed all remarks. Please find below our response and action to each of them. Accordingly, and importantly, we have revised our allocation method. We also included the date when our study was registered with the Iran trial registry and the trial registration number at the end of the Abstract, and we stated Fatemeh Erfanian's contribution in the Authors' Contributions section. All changes are marked in the text.

We trust that the resubmitted paper meets all points made by you and the reviewer and hope that it is suitable for publication in its new form.

Sincerely,

A. Aslani, also on behalf of my co-authors.